Author's response to reviews

Title: Epidemiological Characteristics of Acute Disseminated Encephalomyelitis in Nanchang China: A Retrospective Study.

Authors:

Chang-hui Xiong (xch964595@126.com)
Yan Yan (yanyan@csu.edu.cn)
Zhen Liao (423558826@qq.com)
Shi-hui Peng (48840052@qq.com)
Hai-rong Wen (1076786031@qq.com)
Yan-xia Zhang (309460191@qq.com)
Shu-hua Chen (dhcdcmgk@126.com)
Jian Li (37796884@qq.com)
Hai-ying Chen (chy@nccdc.org.cn)
Xiao-wu Feng (fxw@nccdc.org.cn)
Hui-qing Yao (690207475@qq.com)
Li Huang (364819691@qq.com)
Li Zhang (769300425@qq.com)

Version: 6 Date: 13 August 2013

Author's response to reviews: see over
Cover letter

Dear Rita:

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “Population-based Epidemiologic Study of Acute Disseminated Encephalomyelitis in Nanchang China”. (Manuscript ID: 1020370900102877).

We have studied reviewer’s comments carefully and have made revision which marked in red in the paper. The main corrections in the paper are as following:

1) The manuscript title was revised as “Epidemiological Characteristics of Acute Disseminated Encephalomyelitis in Nanchang China: A Retrospective Study.”
2) Some contents was added in the background section of abstract.
3) Competing interests were explained.
4) Authors' contributions were stated.
5) The quality of written English was improved.
6) The reference number of ethics committee approval was appended

We have tried our best to revise our manuscript according to the comments. Attached please find the revised version, which we would like to submit for your kind consideration.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Corresponding author:
Name: Yan Yan
E-mail: yanyan@csu.edu.cn